Author's response to reviews

Title: The use of opioids at the end of life: a survey of experienced barriers in pain management and knowledge level of physicians

Authors:

Mette L Rurup (m.rurup@vumc.nl)
Christiaan A Rhodius (crhodius@gmail.com)
Manon SA Boddaert (mboddaert@Amstelring.nl)
Astrid GM Keijser (astrid.keijser@gmail.com)
Sander D Borgsteede (s.borgsteede@gmail.com)
H Roeline W Pasman (hrw.pasman@vumc.nl)
Bregje D Onwuteaka-Philipsen (b.philipsen@vumc.nl)

Version: 2 Date: 16 April 2010

Author's response to reviews:

We have added information about medical ethical review